14.1. 1 CS150 Newton

Outline

Last time:

VLSI Chips

This lecture:

Where to from here?

The responsibilities of a technologist

CS150: once over lightly

Relationship to other courses in EECS

14.1. 2

CS150 Newton

"I think there is a world market for maybe five computers"

˙ Thomas Watson, Chairman of IBM, 1943

14.1. 3

CS150 Newton

"There is no reason anyone would want a computer in their home."

˙ Ken Olson, President & Founder, Digital

14.1. 4

CS150 Newton

"Computers in the future may weigh no more than than 1.5 tons."

˙ Popular Mechanics, 1949

14.1. 5

CS150 Newton

"I have traveled the length and breadth of this country and have talked with the best people, and I can assure you that data processing is a fad that won’t last out the year."

˙ The editor in charge of business books for Prentice-Hall, 1957

14.1. 6

CS150 Newton

"But what... is it good for?"

˙ Engineer at Advanced Computing

Systems Division of IBM, 1968, commenting on the role of the microchip.

14.1. 7

CS150 Newton

Semiconductor Industry Growth

Steady State?

1995 2030

Semiconductor as % of Electronics 17% 35%

Electronics as % of GWP 4% 8%

Semiconductors as % of GWP 0.7% 3%

CMOS Technology 0.35 m 0.05 m

World Semiconductor Sales $140B $12,000B

Annual Growth Rate 16% 8%, same as GWP

Source: